Author's response to reviews

Title: Magnitude and concurrence of anxiety, depression among attendees with multiple sclerosis at a tertiary care Hospital in Oman

Authors:

Abdullah Al-Asmi (alasm@squ.edu.om)
Salim Al-Rawahi (salrawahi99@gmail.com)
Zahir Seif Al-Moqbali (moqbali33@gmail.com)
Yahya Al-Farsi (ymfarsi@squ.edu.om)
Musthafa M Essa (drmdessa@gmail.com)
May El- Bouri (may.bouri@gmail.com)
Roopa P Koshy (roopakoshy08@gmail.com)
Arunodaya R Gujjar (grarunoday@hotmail.com)
PC Jacob (jacobpc@squ.edu.om)
Abeer Al-Hodar (abeera@squ.edu.om)
Samir Al-Adawi (samir.al-adawi@fulbrightmail.org)

Version: 2
Date: 16 January 2015

Author's response to reviews:

NA